Author’s response to reviews

Title: Monitoring vaccine and non-vaccine HPV type prevalence in the post-vaccination era in women living in the Basilicata region, Italy

Authors:

Carozzi Francesca (f.carozzi@ispo.toscana.it)
Donella Pulitia (d.puliti@ispo.toscana.it)
Cristina Ocello (c.ocello@ispo.toscana.it)
Pasquale Anastasio (silvio.anastasio@rete.basilicata.it)
Espedito Moliterni (espeditomoliterni@tiscali.it)
Emilia Perinetti (eperinetti@spmsd.com)
Elena Burroni (e.burroni@ispo.toscana.it)
Laurence Serradell (laurence_serradell@yahoo.fr)
Massimo Confortini (m.confortini@ispo.toscana.it)
Paola Mantellini (p.mantellini@ispo.toscana.it)
Marco Zappa (m.zappa@ispo.toscana.it)
Géraldine Dominiak-Felden (gfelden79@yahoo.fr)

Version: 3 Date: 03 Jan 2018

Author’s response to reviews:

Géraldine Dominiak-Felden, MD, MSc

162, Avenue Jean Jaurès
CS 50712
69367 Lyon Cedex 07

Telephone: 04-37-28-41-48
geraldine.dominiak@merck.com
Dear Dr. Harris,

Thank you for accepting for publication our manuscript, Monitoring vaccine and non-vaccine HPV type prevalence in the post-vaccination era in women living in the Basilicata region, Italy. We have revised the manuscript as directed.

Specifically,

1) We moved the Keywords to after the Abstract
2) We replaced “Materials and Methods” with “Methods”
3) We have removed all track changes and highlighting

Please do not hesitate to contact me if you have any questions or need anything further.

Sincerely,

Géraldine Dominiak-Felden, MD, MSc